Author's response to reviews

Title: A Community Participator Approach to Improving Health in a Hispanic Population

Authors:

Michael F Dulin (michael.dulin@carolinashealthcare.org)
Hazel Tapp (hazel.tapp@carolinashealthcare.org)
Heather A Smith (heatsmit@uncc.edu)
Brisa Urquiesta De Hernandez (Brisa.Hernandez@carolinashealthcare.org)
Owen J Furuseth (ojfuruse@uncc.edu)

Version: 7 Date: 10 March 2011

Author's response to reviews:

To answer the one copy edit question we included "or condition" in the text so we could keep HIV/AIDS. Thanks, Hazel Tapp